

**Author's response to reviews**

**Title:** The R.I.R.S. Scoring System: An Innovative Scoring System for Predicting Stone-free Rate Following Retrograde Intrarenal Surgery

**Authors:**

Yinglong Xiao (jimmyxiao201@hotmail.com)

Deng Li (lideng1991@hotmail.com)

Lei Chen (drchenlei@126.com)

Yaoting Xu (ytxu64@vip.163.com)

Dingguo Zhang (jdzdguo@163.com)

Yi Shao (drshaoyi@163.com)

Jun Lu (842320@163.com)

**Version:** 1 **Date:** 25 Sep 2017

**Author's response to reviews:**

The authors' response letter has been included as a supplementary file